Author's response to reviews

Title: Knowledge and Awareness about STDs of Women in Bangladesh

Authors:

Md. Mosharaf Hossain Mr (m_population@yahoo.com)
Kulanthayan Dr K.C. Mani Mr. (kulan@upm.edu.my)
Sherina Dr Mohd Sidik (sherina@upm.edu.my)
Hayati Dr Kadir (hayatik@upm.edu.my)
Md. Rafiqul Dr Islam Mr (rafique_pops@yahoo.com)

Version: 8 Date: 29 June 2014

Author's response to reviews: see over
1. Funding clarification
In the ethics statement you say that "International credit finance (ICF) provided financial and technical assistance for the survey....". This does not appear in the Acknowledgements section. If ICF is a commercial organization this funding need to be declared in the Competing Interests section. Please respond, and correct the manuscript accordingly.

**Answer:** Thanks for your comments. Done, please see the page 14 and line 15 acknowledgements section. ICF is no commercial organization.

2. Additional files

Please DO NOT upload responses to referees and editors as additional files. Use the appropriate tabs in the submission system.

**Answer:** I follow in according to yours comments